Author’s response to reviews

Title: Prevalence and Genetic Characterization of Cryptosporidium, Enterocytozoon, Giardia and Cyclospora in Diarrheal Outpatients in China

Authors:

Hua Liu (liuhua2033@163.com)
Yujuan Shen (amyshyj12@163.com)
Jianhai Yin (chart2543@163.com)
Zhongying Yuan (yuanzhy0606@aliyun.com)
Yanyan Jiang (jiangyy083@yahoo.cn)
Yuxin Xu (ipdxuyx@163.com)
Wei Pan (panwei525@126.com)
Yuan Hu (huyuan7576@sina.com)
Jianping Cao (caojp@yahoo.com)

Version: 5 Date: 31 December 2013

Author’s response to reviews: see over
Dear editor,

We would like to thank you very much for everything you’ve done for the manuscript. We address your comments point by point below:

(I am afraid there might be a problem with the lines number because of the computer software version, the line number of submitted PDF format maybe different from my WORD format. Could you please note it.)

Editor’s comment:

COMMENT 1: line 171 change "infection rates" to "detection rates"
RESPONSE: "infection rates" has been revised as suggested.

COMMENT 1: line 176 change "found to be infected with" to "identified as"
RESPONSE: "found to be infected with" has been revised as suggested.

COMMENT 1: line 178 change "infection rates" to "detection rates"
RESPONSE: "infection rates" has been revised as suggested.

COMMENT 1: line 193 change "identified as important etiologic agents" to "possible etiologic agents"
RESPONSE: "identified as important etiologic agents" has been revised as suggested.

COMMENT 1: line 213 change "infection rates" to "detection rates" and also "were largely due" to "may have been due"
RESPONSE: "infection rates" and "were largely due" have been revised as suggested.
COMMENT 1: line 220 change "infected with" to detected with"
RESPONSE: "infected with" has been revised as suggested.

COMMENT 1: line 233 change "infected with" to "detected with"
RESPONSE: "infected with" has been revised as suggested.

COMMENT 1: line 250 change "intestinal parasitic infections were" to "intestinal parasites were"
RESPONSE: "intestinal parasitic infections were" has been revised as suggested.

Best regards,

Dr. Jianping Cao